Author's response to reviews

Title: Evaluation of Computerized Health Management Information System for Primary Health Care in rural India

Authors:

ANAND KRISHNAN (kanandiyer@yahoo.com)
BARIDALYNE NONGKYNRIH (baridalyne@gmail.com)
KAPIL YADAV (dr_kapilyadav@yahoo.co.in)
SATYAVIR SINGH (satyavir@medialabasia.in)
VIVEK GUPTA (drvivek.gupta@gmail.com)

Version: 4 Date: 10 April 2010

Author's response to reviews: see over
Dated: 10th April 2010

To

The Editor

BMC Public Health

Reference: MS: 1250415418367419

Research article: Evaluation of Computerized Health Management Information System for Primary Health Care in rural India

Dear Sir

Thank you for your mail and for giving me the opportunity to submit a revised version of the manuscript.

You have mentioned that the methodology does not adhere to the RATS guidelines for qualitative research. I have made the necessary changes (highlighted in red in the manuscript).

I am also sending responses and explanation of the paper separately according to the RATS format.

Hoping for a favourable response

Thanking you

Dr Baridalyne Nongkynrih